Author's response to reviews

Title: Factors associated with maternal anaemia among pregnant women in Dhaka

Authors:

Hasina Akhter Chowdhury (hachowdhury01@gmail.com)
Kazi Rumana Ahmed (lopa_birdem@yahoo.com)
Fatema Jebunnesa (runabd2005@yahoo.com)
Jesmin Akter (uttaron_007@yahoo.mail)
Sharmin Hossain (shormybd@yahoo.com)
Md Shahjahan (mdshahjahan@agnionline.com)

Version: 15 Date: 31 July 2015

Author's response to reviews: see over
31 July 2015
To
The Editor
BM C W omen’s Heal th

Subject: Submission of Modified Article
Dear Sir,
We undersigned the authors of the article entitled 'Factors associated with anaemia among pregnant women in Dhaka city' modified the article according to your comments and submitting for your kind consideration and necessary action. In this regard, we also prepared reply which is given below.

Kind regards
Hasina Akhter Chowdhury
Lecturer
Department of Biostatistics
Bangladesh University of Health Sciences (BUHS)
and
Author for correspondence
Em ail: hachowdhury01@gmail.com; lima_125@yahoo.com
Editorial Requirement:

1. Please rename your ‘Introduction’ section into ‘Background’ section.

Reply: Revised the manuscript according to your suggestion.

2. Please include a list of abbreviations that were used on the manuscript. This should follow in the conclusion section of the main article.

Reply: Include a list of abbreviations in the conclusion section of the main article according to your suggestion.